Author's response to reviews

Title: High prevalence of cardiovascular risk factors in children and adolescents with Williams-Beuren syndrome

Authors:

Daiji Takeuchi (0220teamdaiji@gmail.com)
Michiko Furutani (michi@imcir.twmu.ac.jp)
Yuriko Harada (ytr.koro@gmail.com)
Yoshiyuki Furutani (yoshi@imcir.twmu.ac.jp)
Kei Inai (pinai@hij.twmu.ac.jp)
Toshio Nakanishi (pnakanis@hij.twmu.ac.jp)
Rumiko Matsuoka (rumiko@imcir.twmu.ac.jp)

Version: 4 Date: 31 July 2015

Author's response to reviews: see over
30 July 2015

Dr. Nawsheen Boodhun
Executive Editor
BMC Pediatrics

Dear Dr. Boodhun:

Please find attached the manuscript “High prevalence of cardiovascular risk factors in children and adolescents with Williams-Beuren syndrome” by Daiji Takeuchi et al., which we would like to resubmit for publication to BMC Pediatrics. We greatly appreciate that you and the reviewers think our work is important.

This is a revised version of the previous manuscript (MS: 1039586234135477) based on the kind and useful comments by the reviewers. The questions raised by the reviewers have been answered in the “Response to Reviewers” document and the corrections/modifications are marked in red in the revised manuscript.

We thank you for your time and appreciate your kind consideration of our manuscript.

Sincerely,

Daiji Takeuchi, MD, Ph.D
E-mail: ptakeuci@hij.twmu.ac.jp, 0220teamdaiji@gmail.com
Rumiko Matsuoka, M.D, Ph.D.
E-mail: rumiko@imcir.twmu.ac.jp